6.045J/18.400J: Automata, Computability and Complexity Prof. Ron Rivest

Handout 0: Student Survey

5 Wednesday 2003 Matthew Lepinski

Please fill in all the information and return it to a TA before the end of the class. (If you cannot fill this in now, please return it to the course secretary ASAP.) Please write legibly.

1. Name: 2. Year:

3. Course/Major/University: 4. Email Address: 5. Phone:

6. Taking this class for credit?

7. Do you plan to show up at TA Hours: (circle one)

Frequently Infrequently Never 8. What are the best times (on Monday or Tuesday) for TA Hours? 9. Which of the recitations are impossible for you to attend?

Thursday 10:00am Thursday 1:00pm Thursday 4:00pm

10. Have you taken 6.042?

11. If not, why do you think that you are qualified to take this class?

12. Have you had any other classes you believe to be relevant?

13. Why are you taking this class?

14. Are you satisfying a requirement with this class? If so, what grade do you need?

(Note: No promises....)

0-1